Author's response to reviews

**Title:** A cluster randomised controlled trial of nurse and GP partnership for care of Chronic Obstructive Pulmonary Disease

**Authors:**

Nicholas A Zwar (n.zwar@unsw.edu.au)
Oshana Hermiz (oshana.h@unsw.edu.au)
Iqbal Hasan (i.hasan@unsw.edu.au)
Elizabeth J Comino (e.comino@unsw.edu.au)
Sandy Middleton (sandy.middleton@acu.edu.au)
Sanjyot Vagholkar (Sanjyot.Vagholkar@sswhs.nsw.gov.au)
Guy Marks (g.marks@unsw.edu.au)

**Version:** 4  **Date:** 2 May 2008

**Author's response to reviews:**

A section by the title "Authors' contribution" has been added to the manuscript as advised by Dr Lolu da-Silva.